Check for updates

Ultraviolet A radiation and COVID-19 deaths in the USA with replication studies in England and Italy*

M. Cherrie,¹ T. Clemens,¹ C. Colandrea,¹ Z. Feng,¹ D.J. Webb,² R.B. Weller ¹/₁₀³ and C. Dibben¹

¹School of Geosciences, ²Centre for Cardiovascular Science, ³Centre for Inflammation Research, University of Edinburgh, Edinburgh, UK

Linked Comment: R. McKenzie and B. Liley. Br J Dermatol 2021; 185:246–247.

Summary

Correspondence

Richard Weller. Email: r.weller@ed.ac.uk

Accepted for publication 29 March 2021

Funding sources

M.C. was supported by Health Data Research UK (HDR-5012). C.C. was supported by the European Union's Horizon 2020 research and innovation programme Grant agreement No. 676060.

Conflicts of interest

R.B.W. is a director of Dr Weller Ltd (sunscreen development).

Data Availability Statement

Data and code are presented in the following online repository: https://github.com/markocherrie/ COVID19_UVA

*Plain language summary available online

DOI 10.1111/bjd.20093

Background Understanding factors impacting deaths from COVID-19 is of the highest priority. Seasonal variation in environmental meteorological conditions affects the incidence of many infectious diseases and may also affect COVID-19. Ultraviolet (UV) A (UVA) radiation induces release of cutaneous photolabile nitric oxide (NO) impacting the cardiovascular system and metabolic syndrome, both COVID-19 risk factors. NO also inhibits the replication of SARS-CoV2.

Objectives To investigate the relationship between ambient UVA radiation and COVID-19 deaths.

Methods COVID-19 deaths at the county level, across the USA, were modelled in a zero-inflated negative-binomial model with a random effect for states adjusting for confounding by demographic, socioeconomic and long-term environmental variables. Only those areas where UVB was too low to induce significant cutaneous vitamin D3 synthesis were modelled. We used satellite-derived estimates of UVA, UVB and temperature and relative humidity. Replication models were undertaken using comparable data for England and Italy.

Results The mortality rate ratio (MRR) in the USA falls by 29% [95% confidence interval (CI) 40% to 15%) per 100 kJ m⁻² increase in mean daily UVA. We replicated this in independent studies in Italy and England and estimate a pooled decline in MRR of 32% (95% CI 48% to 12%) per 100 kJ m⁻² across the three studies.

Conclusions Our analysis suggests that higher ambient UVA exposure is associated with lower COVID-19-specific mortality. Further research on the mechanism may indicate novel treatments. Optimized UVA exposure may have population health benefits.

What is already known about this topic?

- Infectious disease often shows seasonality.
- Cutaneous production of nitric oxide (NO) following ultraviolet A (UVA) exposure reduces blood pressure, a cardiovascular risk factor.
- Comorbid cardiovascular disease worsens COVID-19 outcomes.
- NO limits SARS-CoV and SARS-CoV2 replication in vitro.

What does this study add?

- Higher ambient UVA is associated with lower risk of COVID-19 death in three countries, when vitamin D synthesis would have been limited.
- Sun exposure may reduce death from COVID-19 independently of vitamin D.

363

Seasonality¹ and variation in temperature,² humidity³ and ultraviolet (UV) radiation⁴ are related to the incidence of several infectious diseases. In this paper we explore whether UVA might independently affect COVID-19 outcomes. We have previously described a novel nitric oxide (NO)-driven, vitamin D-independent mechanism⁵ by which sunlight can lower blood pressure, and at the population level we have shown that UV is associated with lower blood pressure⁶ and a reduced incidence of myocardial infarctions.⁷ The same UV-driven mechanism may also cause seasonal variation in the development of diabetes and metabolic syndrome.⁸

UV may have a direct effect on the viability of SARS-CoV2 virus in airborne droplets and on fomites.⁹ thus reducing both infection rates and the size of inoculum in those becoming infected, with correspondingly reduced disease severity^{10,11} and COVID-19 growth rates.¹² Direct viricidal suppression of SARS-CoV2 appears to be a UVB effect⁹ with UVA wavelengths having no effect on SARS-CoV1.13 However, UVA does photo-release NO from stores in the skin whence it is mobilized to the systemic circulation, causing vasodilatation and reduction in blood pressure,⁵ offering cardiovascular and metabolic benefits from UV exposure.^{5,8} As cardiometabolic disease and possibly hypertension¹⁴ increase the risk of death from COVID-19, any UV-driven improvements in these risk factors would be expected to reduce mortality.¹⁵ NO may also have a specific effect on COVID-19. It inhibits replication of SARS-CoV¹⁶ and SARS-CoV2.17 In the case of SARS-CoV this occurs by S-nitrosation of the spike protein, preventing the posttranslational palmitoylation required for fusion with its cognate angiotensin-converting enzyme 2 receptor (ACE2R).¹ The spike protein of SARS-CoV is highly homologous to that of SARS-CoV2,^{19,20} suggesting that NO may similarly limit binding to ACE2R by SARS-CoV2, accounting for reduced disease transmission and severity.

Given the apparent greater severity of illness and risk of death from COVID-19 among those with cardiometabolic diseases, ^{21,22} the importance of season for infectious diseases, and a plausible pathway to reduced disease transmission and severity through photo-released NO, we investigate whether ambient UVA exposure is associated with COVID-19 deaths across the USA, independent of other UV pathways, and whether the finding is replicated in studies of England and Italy.

Methods

Study setting and participants

We used an ecological regression approach to model COVID-19 deaths in small areas (counties) across the contiguous USA during the early part of the COVID-19 pandemic (January to April 2020). Our main analysis was for USA counties (n = 2474) with replication studies for COVID-19 deaths across English 'middle layer super output areas' (MSOAs) (n = 6724) and excess deaths across Italian municipalities (n = 6775). To reduce potential confounding through a UVB vitamin D pathway [a monthly mean UV radiation on the 252–330 nm spectrum (action spectrum of UV radiation for vitamin D production, UV_{vitD} , of under 165 kJ m⁻²)²³], we included only 'small areas' that were experiencing levels of UV too low to induce significant cutaneous vitamin D3 synthesis at any time during the study period ('UV vitamin D winter'). This meant that 2474 counties (out of 3088) in the USA were included within the analysis shown in Figure 1.

COVID-19 deaths

USA COVID-19 deaths between 22 January and 30 April 2020 came from data compiled by the Center for Systems Science and Engineering at Johns Hopkins University.²⁴ A COVID-19 death was defined as a case if the practitioner suspected that COVID-19 played a role, even if it was not directly attributable to the death. English COVID-19 deaths between 1 March and 17 April 2020 were drawn from data compiled by the UK Office for National Statistics (ONS).²⁵ Deaths were included if COVID-19 was mentioned on the death certificate, with a delay of usually 5 days between occurrence and registration. In Italy there is no COVID-19-classified mortality data available for municipalities. Instead, we estimated this from excess deaths drawn from ISTAT (Italian Institute of Statistics)²⁶ available for 7270 of 7904 municipalities for 1 March to 30 April for 2015–2019 compared with 2020.

Ambient ultraviolet radiation

We derived mean daily UVA for the small areas in each study for these date ranges: USA (1 January - 30 April); England (1 January - 17 April) and Italy (1 January - 30 April). We start our time series of UVA before the period in which we were recording deaths because of the lag between infection and death. The UVA dataset was produced by JAXA (Japan Aerospace Exploration Agency) using the MODerate resolution Imaging Spectroradiometer (MODIS) instrument on board NASA's Aqua and Terra satellites.²⁷ Atmospheric absorption and scattering due to ozone and water (vapour and cloud) were accounted for by using a simplified planetary atmosphere (clear atmosphere positioned above a cloud layer). Downward irradiance values (i.e. combined direct and diffuse radiation on a horizontal plane) for UVA (315-400 nm) were converted to daily values by using the diurnal cycle of solar zenith angle with instantaneous atmospheric conditions. UVA data were aggregated for USA counties, English MSOAs and Italian municipalities and expressed as mean daily kJ m⁻².

A long-term UV_{vitD} dataset (30-year monthly average) developed by the National Center for Atmospheric Research (NCAR) was used.²⁸ UV_{vitD} data were aggregated and expressed in mean monthly kJ m⁻². We used the highest quintile as the cutoff for year-round vitamin D synthesis, which corresponds to a monthly mean of over 165 kJ m⁻².²⁹



Figure 1 Average daily mean UVA (kJ m⁻²) January to April: (a) the USA, (b) England and (c) Italy. The UVA colour scale differs between countries. USA counties shown in grey were excluded from the study because they had monthly mean UV_{vitD} of over 165 kJ m⁻².

Covariates

Demographic and socioeconomic

A number of demographic, socioeconomic, long-term environmental exposure and infection susceptibility variables were used in our models. This was to appropriately adjust for spatial associations, with both UVA and COVID-19 mortality, which might otherwise lead to a spurious relationship between UVA and COVID-19 mortality. We identified these from the existing literature and measured them at the smallarea level. Our zero-inflated negative-binomial (ZINB) model provides for different ways that covariates could be included. We assessed for each covariate whether it might (i) be a direct confounder for the UVA-COVID-19 relationship, in which case they were added to the negative-binomial (NB) part of the model and (ii) impact the exposure of individuals to the virus, in which case they were added to the zero-inflated (ZI) part of the model. Covariates might therefore be in both parts of the model.

Older age and ethnicity were associated with higher risk of COVID-19 death, possibly due to higher prevalence of comorbidities, including hypertension, heart disease and respiratory diseases.³⁰ We used country-specific datasets to measure risk factors associated with age and ethnicity.

Poorer citizens are at higher risk of infection due to essential working and death due to pre-existing health conditions.³¹ We used country-specific datasets of socioeconomic deprivation. For the USA we took the first principal component score from a principal component analysis of: percentage in poverty, median house value, median house income, percentage owner-occupied and percentage of population with less than a high-school education. To capture socioeconomic deprivation in England we used percentage of residents under 21 years of age who did not enter higher education and an incomedeprivation score.³² For Italy, we used the Italian Deprivation Index.³³

Long-term environment

Higher $PM_{2.5}$ (fine particles with diameter $\leq 2.5 \ \mu m$) exposure is linked with a range of respiratory and cardiovascular diseases and has been shown to increase COVID-19 mortality rate in other analyses.³⁴ Long-term PM_{2.5} (2000–2016) data at a $0.01^{\circ} \times 0.01^{\circ}$ resolution were modelled using satellite and monitored $PM_{2.5}$ station data.³⁵ We used these data for both the USA and Italy. In England, long-term 2014-2018 $PM_{2\cdot 5}$ at a 1 km \times 1 km resolution was modelled using monitored PM_{2.5} station data.³⁶ Variation in temperature is associated with COVID-19 mortality.³⁷ Long-term mean monthly winter temperature (December–February) at a 4 km \times 4 km resolution for 2000–2016 was modelled using satellite data³⁸ for the USA. Long-term mean monthly winter temperature (December-February) at a 1 km \times 1 km resolution for 1981-2010 was modelled using interpolation of Meteorological Office weather stations for England.³⁹ Long-term median land surface temperature (December-February), daytime monthly median value at a 1 km \times 1 km resolution for 2000–2017, was modelled using satellite data for Italy.⁴⁰

Viral exposure

The population 'at-risk' needs to be adjusted for exposure to the virus in case factors increasing or decreasing risk of

published by John Wiley & Sons Ltd on behalf of British Association of Dermatologists

exposure are associated with spatial variance in UVA levels. In densely populated, urban or peri-urban areas, with high use of public transport, COVID-19 transmission is faster and the prevalence of cases higher. Probable exposure is therefore estimated through population density, urban/rural status and state percentage of positive COVID-19 tests in the USA. We used population density from the 2018 mid-year population estimates of the UK ONS, percentage of residents using different forms of transport (bus, train, tube) from the 2011 census, and upper tier local authority (UTLA) number of days since a local authority had 10 confirmed cases in England. We used ISTAT population density from 2019, the municipality area, and total cases in province in Italy up to 30 April.²⁶

Statistical analysis

We used a ZINB regression to model counts of death in small geographical areas because the counts of deaths were likely to be spatially variable with the mean counts of deaths across small areas likely to be much smaller than the variance of the counts between the small areas and an excess of zeros (areas with no COVID-19 infection).

We included a random effect in the model. This was a random intercept for a higher and administratively important geographical unit in each country. In the USA this was the state, in England the local authority and in Italy the province in order to: (i) capture the systematic way the risk of death from COVID-19 might be related to this higher geography (e.g. due to differences in political administration, health services, funding, public health effects or levels of infection) and (ii) incorporate spatial clustering in the estimation of standard errors. The linearity of the relationship between UVA and COVID-19 deaths was tested using fractional polynomial regression, which tests whether fractional powers (nonlinear) improve the fit of the model. The linear model was found to be the best fit for each country. Each of the country models were specified independently by separate team members. All models were fitted using the glmmTMB package for R,⁴¹ which fits random-effect generalized linear models described in Bolker et al.⁴²

We carried out a meta-analysis to estimate the pooled effect across the three studies using a random-effects model. We used a restricted maximum-likelihood estimator with no adjustments.

Software

All analyses were undertaken in R 3.6.1 (https://www.r-pro ject.org/). We predicted the number of deaths per million population at suitable levels of UVA by calculating the marginal means in the 'emmeans' package in R^{43} We used a random-effects model as part of the 'metafor' package in R^{44} to calculate a cross-county pooled estimate of the mortality rate ratio (MRR).

Data and code

Data and code are presented in the following online repository: https://github.com/markocherrie/COVID19_UVA

Results

Daily mean UVA (January–April 2020) varied between 340 and 1000 kJ m⁻² across the three countries, with lower average levels experienced across England (340–460 kJ m⁻²) during the period compared with Italy (600–900 kJ m⁻²) and USA (500–1000 kJ m⁻²) (Figure 1a–c). For the USA the average county UVA level between January and 30 April was 696 kJ m⁻² (SD 83). For England the average MSOA UVA level between January and 17 April was 412 kJ m⁻² (SD 18). For Italy the average UVA level between January and 30 April was 717 kJ m⁻² (SD 52).

We model the rate of COVID-19 deaths in small areas in a multilevel ZINB model for the USA, England and Italy (Table 1). The model estimate is the MRR per 100 (kJ m⁻²) increase in UVA or the ratio of the mortality rate in a small area with a similar area exposed to 100 (kJ m⁻²) more UVA. Adjusting for other confounders measured at the small-area level (county, MSOA and municipality) and adjusting for infection rates at a higher level of geography (state, UTLA and province) (Table 2) our estimates show reductions in MRR of 0.71 in the USA per 100 kJ m⁻² increase in UVA (Figure 2). We found a similar size of effect in our two replication studies: an MRR in Italy of 0.81 and in England 0.51. For the random-effects meta-analysis across the three countries we find a pooled MRR estimate of 0.68 per 100 increase in UVA (kJ m⁻²) (Figure 2).

Using the model (including the random effects) we predict what the rate of mortality (deaths per million population) would be for different experienced levels of UVA (all other coefficients in the model held at their means) in the three countries (Figure 3a–c). This illustrates that the models predict a similar approximate fall by a third to a half in the average risk of death across the levels of UVA experienced across the three countries. This is despite the very different levels of death being experienced between the countries at this point in the epidemic. The models also suggest that the size of the risk reduction per 100 (kJ m⁻²) UVA is largest for England which has the lowest average level of UVA during the period.

Discussion

Our analysis finds that higher ambient UVA exposure is associated with lower COVID-19-specific mortality. This effect appears independent of differences in socioeconomic composition, temperature, humidity and UV within the vitamin D action spectrum. Our models are consistent with a situation where UVA exposure may be an additional UV-protective factor against COVID-19, along with other potential UV-related pathways through vitamin D production and direct viricidal suppression. This is an observational study and so further research is required.

Table 1 Zero-inflated negative-binomial models for the USA, England and Italy showing the change in COVID-19 mortality rate ratio per $100 \text{ kJ} \text{ m}^{-2}$ increase in UVA

Model ^a /	Geographic units			Adjusted MRR (95% CI) per 100 kJ m^{-2}	
Country	of analysis	Random effect	Deaths, n	increase in UVA	ICC
USA	Counties $(n = 2474)$	State $(n = 46)$	62 219 (COVID-19 deaths)	0.71 (0.60-0.85)	0.042
England	MSOAs $(n = 6724)$	UTLA $(n = 150)$	19 315 (COVID-19 deaths)	0.51 (0.39–0.66)	0.231
Italy	Municipalities $(n = 6775)$	Province $(n = 104)$	46 095 (excess deaths)	0.81 (0.71–0.93)	0.068

CI, confidence interval; ICC, intraclass correlation coefficient; MRR, mortality rate ratio; MSOA, middle layer super output areas; UTLA, upper tier local authority; UVA, ultraviolet A radiation. ^aAdjusted for the variables shown in Table 2.

Given that UVA does not appear to directly act in viricidal suppression of SARS-CoV2,^{9,13} the reduction in the observed risk might be the result of either behavioural change or a biological pathway. Warm and sunny days might increase the chance of time spent outdoors or away from indoor spaces where close contact with others would be higher. Although UVA and air temperature are associated, in our models we also control for temperature. Our findings are consistent between three countries and with varying latitudes. It seems unlikely that time spent outdoors would be sufficiently linearly associated with UVA for the same calendar period, after controlling for temperature, over such a wide range of latitude and UVA to

produce the same effect on time spent outdoors (e.g. between Newcastle and Naples in March). In contrast, there is increasing evidence that UVA photo-releases NO from skin and that NO has important potential impacts on virus replication.¹⁶⁻¹⁸ NO Snitrosates the spike protein of SARS-CoV and also the SARS-CoV and SARS-CoV2 protease. This double action blocks the myristolation of the spike protein necessary for it to bind to ACE2R, and also inhibits viral protease activity, with consequent reduced viral replication and cytopathic effects.¹⁶⁻¹⁸ Maintenance of the health vasculature is dependent on constitutive endothelial NO synthase-derived NO. Endothelial damage and coagulation may excess underlie widespread organ

Table 2 Models adjusted for these variables

Model / Country	Geographic units of analysis	Negative-binomial model, adjusted for	Zero-inflated model, adjusted for
USA	County level	 PM_{2.5} UV_{vitD} Winter temperature Winter humidity Percentage of residents: 65+ years, black, Hispanic Deprivation score Urban/rural 	 Percentage of residents: 65+ years, black, Hispanic Deprivation score Urban/rural
	State level	• Proportion of positive COVID-19 cases	• Proportion of positive COVID-19 cases
England	MSOA level	 PM_{2.5} Long-term winter temperature Percentage of residents: 80+ years, 65–79 years, black, Indian, Pakistani/Bangladeshi, Chinese, in care homes, in higher education Income-deprivation score 	 Percentage of residents: 80+ years, 65–79 years, black, Indian, Pakistani/Bangladeshi, Chinese, in care homes, in higher education, using public transport (bus, train, tube) Income-deprivation score Population density
	UTLA level	• Number of days since a local authority had 10 confirmed cases	• Number of days since a local authority had 10 confirmed cases
Italy	Municipalities level	 PM_{2.5} Long-term winter temperature Number of foreign born Percentage of residents: 65+ years, 85+ years Population density Municipality area Deprivation score 	 Number of foreign born Percentage of residents: 65+ years, 85+ years Population density Municipality area Deprivation score
	Province level	• Total cases in province	• Total cases in province

MSOAs, middle layer super output areas; $PM_{2.5}$, fine particles with diameter $\leq 2.5 \ \mu m$; UTLA, upper tier local authority.



Figure 2 Mortality rate ratios per 100 kJ m^{-2} increase in mean daily UVA [pooled estimate from random effects (RE) model].

involvement⁴⁵ in COVID-19, which would be mitigated by photochemical NO production. Although further research is required, we suggest that this NO biological pathway is a plausible explanation for our model results. UVA also correlates inversely with population blood pressure and cardiovascular disease^{6,7} and associates with reduced all-cause mortality, largely through a reduction in cardiovascular deaths.⁴⁶

Tolerance to UVA, as shown by the reduced fall in COVID-19 deaths we demonstrated for a given incremental rise in UVA at higher irradiances in the USA and Italy, compared with England, could be explained by increasing melanin in the skin blocking UV penetration. Adaptive pigmentation occurs even at relatively low levels of UVA, with approximately half of the annual variation in forehead pigmentation for white-skinned individuals living in Copenhagen (55.68° N) occurring between January and April.⁴⁷

Deaths from COVID-19 are disproportionately high in people of black African or Asian ancestry in Europe and the USA.^{48,49} Social factors probably account for much of this, but we have shown that the vitamin D and temperatureindependent fall in blood pressure with increased UV is attenuated in black compared with white Americans.⁶ White skin colour is an evolutionary adaptation that occurred around 5000 years ago in Neolithic agriculturalists migrating to lowlight northern Europe.⁵⁰ Darker skin colour would thus be anticipated to reduce any biological benefits from UV, whether blood pressure reduction or infection prevention, particularly at higher latitudes. The remarkably low COVID-19 mortality in equatorial Africa is consistent with this.⁵¹

There are of course weaknesses in our design. We have adjusted our models for all the clinically significant factors that we believe might be spatially or temporally associated with both UVA and COVID-19 mortality risk but if any unmeasured factors exist, they might plausibly explain the relationships identified. UVA and covariates are measured at the small-area level not in individuals. We therefore assume the unmeasured individual-level model is replicated in the ecological model and this may not be the case. However, it is unclear what relationship between UVA and different geographical areas could exist that would modify its relationship with COVID-19. Ambient UVA is likely to be a poor measure of individual-level UVA level exposure (e.g. exposure could be mainly derived from indoor environments). However, in an ecological model the risk associated with personal exposure is averaged across all people in a small area and, therefore, the 'mismeasurement' of personal UVA exposure is predominantly Berkson error. This type of error will not bias an ecological model, unless the error is associated, at the small-area level, with ambient UVA level. For this to systematically occur in our models seems unlikely. Satellite-derived UVA measures may be overestimated under high aerosol loadings (e.g. when the PM2.5 count is



Figure 3 Predicted COVID-19 rates of deaths at selected levels of UVA in (a) the USA, (b) England and (c) Italy, given the model random effect, at the mean level of all other covariates. The predicted risks reflect the pandemic situation (i.e. infection levels) in each country at the time of the study.

high) and this may mean that our model, because we adjust for $PM_{2\cdot5}$, underestimates the association between actual UVA exposure and COVID-19 deaths. There could be misclassification of deaths and rates of infection are estimated within the model and with indirect measures. However, any resulting measurement errors seem unlikely to be correlated with spatial variation in UVA and therefore biasing. The random effect in our models will incorporate differences between socially and politically distinct regions (states, local authorities and municipalities) that might induce a spurious relationship between UVA and mortality. The replication of the findings across three countries with very different health systems, economic and political structures, pandemic situations and climates suggests a robust finding.

In conclusion, this study is observational and therefore any causal interpretation needs to be taken with caution. However, if the relationship identified proves to be causal, it suggests that optimizing sun exposure may be a possible public health intervention. Given that the effect appears independent of a vitamin D pathway, it suggests possible new COVID-19 therapies and the importance of exploring the role of circulating NO.

Acknowledgments

Thank you Hiwot Weller for filming the video summary.

References

- 1 Fisman D. Seasonality of viral infections: mechanisms and unknowns. Clin Microbiol Infect 2012; **18**:946–54.
- 2 Earn DJ, He D, Loeb MB et al. Effects of school closure on incidence of pandemic influenza in Alberta, Canada. Ann Intern Med 2012; 156:173-81.
- 3 Shaman J, Pitzer VE, Viboud C et al. Absolute humidity and the seasonal onset of influenza in the continental United States. PLoS Biol 2010; **8**:e1000316.
- 4 Charland KM, Buckeridge DL, Sturtevant JL et al. Effect of environmental factors on the spatio-temporal patterns of influenza spread. Epidemiol Infect 2009; 137:1377–87.
- 5 Liu D, Fernandez BO, Hamilton A et al. UVA irradiation of human skin vasodilates arterial vasculature and lowers blood pressure independently of nitric oxide synthase. J Invest Dermatol 2014; 134:1839–46.
- 6 Weller RB, Wang Y, He J et al. Does incident solar ultraviolet radiation lower blood pressure? J Am Heart Assoc 2020; 9:e013837.
- 7 Mackay DF, Clemens TL, Hastie CE et al. UVA and seasonal patterning of 56 370 myocardial infarctions across Scotland, 2000–2011. J Am Heart Assoc 2019; 8:e012551.
- 8 Geldenhuys S, Hart PH, Endersby R et al. Ultraviolet radiation suppresses obesity and symptoms of metabolic syndrome independently of vitamin D in mice fed a high-fat diet. Diabetes 2014; 63:3759–69.
- 9 Ratnesar-Shumate S, Williams G, Green B et al. Simulated sunlight rapidly inactivates SARS-CoV-2 on surfaces. J Infect Dis 2020; 222:214–22.

- 10 Paulo AC, Correia-Neves M, Domingos T et al. Influenza infectious dose may explain the high mortality of the second and third wave of 1918–1919 influenza pandemic. PLOS ONE 2010; 5:e11655.
- 11 Yu X, Sun S, Shi Y et al. SARS-CoV-2 viral load in sputum correlates with risk of COVID-19 progression. Crit Care 2020; 24:170.
- 12 Carleton T, Cornetet J, Huybers P et al. Global evidence for ultraviolet radiation decreasing COVID-19 growth rates. Proc Natl Acad Sci 2021; 118:e2012370118.
- 13 Darnell ME, Subbarao K, Feinstone SM, Taylor DR. Inactivation of the coronavirus that induces severe acute respiratory syndrome, SARS-CoV. J Virol Methods 2004; 121:85–91.
- 14 Gao C, Cai Y, Zhang K et al. Association of hypertension and antihypertensive treatment with COVID-19 mortality: a retrospective observational study. Eur Heart J 2020; 41:2058–2066.
- 15 Lindqvist PG. The winding path towards an inverse relationship between sun exposure and all-cause mortality. Anticancer Res 2018; 38:1173–8.
- 16 Akerström S, Mousavi-Jazi M, Klingström J et al. Nitric oxide inhibits the replication cycle of severe acute respiratory syndrome coronavirus. J Virol 2005; 79:1966–9.
- 17 Akaberi D, Krambrich J, Ling J et al. Mitigation of the replication of SARS-CoV-2 by nitric oxide in vitro. Redox Biol 2020; 37:101734.
- 18 Akerström S, Gunalan V, Keng CT et al. Dual effect of nitric oxide on SARS-CoV replication: viral RNA production and palmitoylation of the S protein are affected. Virology 2009; 395:1–9.
- 19 Grifoni A, Sidney J, Zhang Y et al. A sequence homology and bioinformatic approach can predict candidate targets for immune responses to SARS-CoV-2. Cell Host Microbe 2020; 27:671–680.e2.
- 20 Wrapp D, Wang N, Corbett KS et al. Cryo-EM structure of the 2019-nCoV spike in the prefusion conformation. Science 2020; **367**:1260–3.
- 21 Wu Z, McGoogan JM. Characteristics of and important lessons from the coronavirus disease 2019 (COVID-19) outbreak in China: summary of a report of 72314 cases from the Chinese Center for Disease Control and Prevention. JAMA 2020; **323**:1239–42.
- 22 Guan W-J, Liang W-H, Zhao Y et al.; on behalf of the China Medical Treatment Expert Group for COVID-19. Comorbidity and its impact on 1590 patients with COVID-19 in China: a nationwide analysis. Eur Respir J 2020; 55:2000547.
- 23 Bouillon R, Eisman J, Garabedian M et al, International Commission on Illumination (CIE). Action Spectrum for the Production of Previtamin D3 in Human Skin; Technical Report 174. Vienna: CIE, 2006.
- 24 Dong E, Du H, Gardner L. An interactive web-based dashboard to track COVID-19 in real time. Lancet Infect Dis 2020; **20**:533-34.
- 25 Office for National Statistics. Deaths Involving COVID-19 by Local Area and Socioeconomic Deprivation: Deaths occurring between 1 March and 17 April 2020. Available at: https://www.ons.gov.uk/peoplepopulationa ndcommunity/birthsdeathsandmarriages/deaths/bulletins/deathsin volvingcovid19bylocalareasanddeprivation/deathsoccurringbetwee n1marchand17april (last accessed 19 April 2021).
- 26 Italian Institute of Statistics [ISTAT]. Deaths and causes of death: what ISTAT produces. 2020. Available at: https://www.istat.it/it/ archivio/240401 (last accessed 19 April 2021).
- 27 Saigusa N, Ichii K, Murakami H et al. Impact of meteorological anomalies in the 2003 summer on Gross Primary Productivity in East Asia. Biogeosciences 2010; **7**:641–55.
- 28 Lee-Taylor J, Madronich S. Climatology of UV-A, UV-B, and Erythemal Radiation at the Earth's Surface, 1979–2000; Technical report no. NCAR/TN-474+STR. Boulder, CO: National Center for Atmospheric Research, 2007. https://doi.org/10.5065/D62J68SZ.
- 29 Holick MF. High prevalence of vitamin D inadequacy and implications for health. Mayo Clinic Proc 2006; 81:353-73.

- 30 Centers for Disease Control and Prevention. People at increased risk and other people who need to take extra precautions. 2020. Available at: https://www.cdc.gov/coronavirus/2019-ncov/needextra-precautions/index.html?CDC_AA_refVal=https%3A%2F%2Fwww. cdc.gov%2Fcoronavirus%2F2019-ncov%2Fneed-extra-precautions% 2Fpeople-at-increased-risk.html (last accessed 19 April 2021).
- 31 Han Y, Li VOK, Lam JCK et al. Who is more susceptible to Covid-19 infection and mortality in the States? mdRxiv 2020. https://doi. org/10.1101/2020.05.01.20087403 (preprint).
- 32 Ministry of Housing, Communities & Local Government. English Indices of Deprivation 2019 (IoD2019). London: Ministry of Housing, Communities and Local Government, 2019.
- 33 Caranci N, Biggeri A, Grisotto L et al. L'indice di deprivazione italiano a livello di sezione di censimento: definizione, descrizione e associazione con la mortalità. Epidemiol Prev 2010; 34:167–76.
- 34 Wu X, Nethery RC, Sabath MB et al. Air pollution and COVID-19 mortality in the United States: strengths and limitations of an ecological regression analysis. Sci Adv 2020; 6:eabd4049.
- 35 Atmospheric Composition Analysis Group. Surface PM_{2.5}. 2020. Available at: http://fizz.phys.dal.ca/~atmos/martin/?page_id=140 (last accessed 19 April 2021).
- 36 Department for Environment Food & Rural Affairs. Modelled background pollution data. 2020. Available at: https://uk-air.defra. gov.uk/data/pcm-data (last accessed 19 April 2021).
- 37 Ma Y, Zhao Y, Liu J et al. Effects of temperature variation and humidity on the death of COVID-19 in Wuhan, China. Sci Total Environ 2020; 724:138226.
- 38 University of Idaho. GRIDMET: University of Idaho Gridded Surface Meteorological Dataset. Earth Engine Data Catalog, 2020. Available at: https://developers.google.com/earth-engine/datase ts/catalog/IDAHO_EPSCOR_GRIDMET (last accessed 19 April 2021).
- 39 Centre for Environmental Data Analysis (CEDA). HadUK-Grid gridded and regional average climate observations for the UK. 2020. Available at: http://data.ceda.ac.uk/badc/ukmo-hadobs/data/ insitu/MOHC/HadOBS/HadUK-Grid/v1.0.1.0/1km/tas/mon-30y/ v20190808 (last accessed 19 April 2021).
- 40 EnvirometriX Ltd. OpenLandMap long-term land surface temperature daytime monthly median. Earth Engine Data Catalog, 2021. Available at: https://developers.google.com/earth-engine/datase

ts/catalog/OpenLandMap_CLM_CLM_LST_MOD11A2-DAY_M_v01 (last accessed 19 April 2021).

- 41 Brooks ME, Kristensen K, van Benthem KJ et al. glmmTMB balances speed and flexibility among packages for zero-inflated generalized linear mixed modeling. R Journal 2017; **9**:378–400.
- 42 Bolker BM, Brooks ME, Clark CJ et al. Generalized linear mixed models: a practical guide for ecology and evolution. Trends Ecol Evol 2009; 24:127–35.
- 43 Lenth RV, Buerkner P, Herve M et al. emmeans: Estimated marginal means, aka least-squares means. Available at: https://cran.rproject.org/web/packages/emmeans/index.html (last accessed 19 April 2021).
- 44 Viechtbauer W. Meta-analysis package for R (metafor). 2020. Available at: http://www.metafor-project.org;https://github.com/ wviechtb/metafor (last accessed 19 April 2021).
- 45 Teuwen LA, Geldhof V, Pasut A, Carmeliet P. COVID-19: the vasculature unleashed. Nat Rev Immunol 2020; 20:389–91.
- 46 Alfredsson L, Armstrong BK, Butterfield DA et al. Insufficient sun exposure has become a real public health problem. Int J Environ Res Public Health 2020; 17:5014.
- 47 Lock-Andersen J, Wulf HC. Seasonal variation of skin pigmentation. Acta Derm Venereol 1997; 77:219–21.
- 48 Chen Y-H, Glymour MM, Catalano R et al. Excess mortality in California during the coronavirus disease 2019 pandemic, March to August 2020. JAMA Intern Med 2021; 181:705–7.
- 49 Chin-Hong P, Alexander KM, Haynes N et al. Pulling at the heart: COVID-19, race/ethnicity and ongoing disparities. Nat Rev Cardiol 2020; 17:533-5.
- 50 Skoglund P, Mathieson I. Ancient genomics of modern humans: the first decade. Annu Rev Genomics Hum Genet 2018; **19**:381–404.
- 51 Uyoga S, Adetifa IMO, Karanja HK et al. Seroprevalence of anti-SARS-CoV-2 IgG antibodies in Kenyan blood donors. Science 2021; 371:79–82.

Supporting Information

Additional Supporting Information may be found in the online version of this article at the publisher's website:

Powerpoint S1 Journal Club Slide Set.